MEG Management Committee meeting

21.01.2009, 14:00hrs, West Wing Seminar Room, CBU

Minutes
· Present:  Friedemann Pulvermuller, Yury Shtyrov, Gary Chandler, Clare Dine, Maarten Van Casteren, Rik Henson, Oleg Korzyukov, Olaf Hauk, Marie Smith, Georgie Morrill.

1. Apologies for absences
· Rob Anthony

· Rhodri Cusack

2. Matters arising 
None
3. Minutes of the last meeting
The minutes were approved.
4. Computing








YS reported that the acquisition software has been updated again so asks all to look out for any error messages/changes and to report to Neuromag if there are any issues.  MvC mentioned that when fitting the HPI coils, the system gives you a ‘confidence percentage’ which should be higher than 98% but has in the past been as low as 40 or 50% when something was wrong.  On two occasions recently MvC has encountered percentages as low as -300% which means there is a problem somewhere.  MS agreed that she has had the same problem.  FP suggested that this should be an urgent problem which should be fixed straight away.
Action: MvC to email errors to Neuromag to ask advice.
5. Lab business and progress
5.1 Annual Service over Christmas

YS reported that everything seems to be okay since the service and is running a lot quieter.  Neuromag fixed the broken channels that we knew of plus some that we weren’t aware were broken.  The next service will probably be the same time this year. 

5.2 Wall for sound insulation

YS showed the committee the drawings for the new wall in the MEG lab. FP was concerned about the clashing of doors and MvC suggested swapping the doors into the machine around so that they can open easier. GC/YS will be liaising with Neuromag to sort this.  YS also informed that Neuromag have offered to redo the flooring, decorating of walls and ceiling.

Action:  YS/GC liaise with Neuromag about changes to drawings.
5.3 Review of the MEG admin – Log/screening sheets

CD reported that she had looked at the SOPs and it states that the form must be used before the participant goes into the MEG machine.  FP suggested liaising with Marion to check whether Ethics approval is needed to change the form.  RH thinks that you only need ethics approval for individual studies where any changes in the form can be mentioned.  CD informed the committee that the SOPs were very out of date, so FP suggested that CD go through them, make any changes and send them to him once done.

Action: CD to speak to Marion about possibilities to update SOPs, CD would also update SOPs and screening sheet and send it to FP for double checking.
5.4 Other issues
Eye tracker:  YS says that the Eye tracker should be added to the agenda again.  MvC tried it but couldn’t make it work.  YS suggested the MvC keep trying and then report to Neuromag with any issues.  GC also suggested that he can always ask Rich to come back in a do some more training.

Visual Supine:  YS brought up that there was talk of getting a mirror.  GC reported that Aidan tried with EEG but it wasn’t very comfortable. RH suggested using some padding or something. OK offered that he ask how they do it in Detroit. 
6. New Study design:
Recording of "Gold Standard" MEG/ERP data sets for comparisons across different data analyses algorithms.  OK proposed a study including 9 conditions, to be run with 5 subjects. He would analyse MEG responses using different software and localisation methods to compare the results. This would be a methods project involving a range of MEG scientists. After discussion it was agreed that the experiment was, in principle, a good idea. Further suggestions were also made, for example that there should be some self-paced button press recording within this study. It was also mentioned that features of the study would need to be spelled out in more detail (number of stimuli, exact time required, reason for the many conditions, hypotheses).
Action: OK to finalise the design in collaboration with relevant MEG scientists
7. MEG lab for January
GM reported that it had been quiet apart from the GSK slots but the lab was slowly filling up.  GSK are struggling to get controls, let alone patients so have cancelled a few slots recently. They have reduced the amount of participants from 27 down to 15 which they will test twice.  FP reported that the deadline internally is October 2009.
AOB: 
OH would like to run some ‘phantom measurements’ where he will be able to look at the analysing with certain parameters to see what occurs.  The committee agreed with was a good idea.

9. Date of the next meeting:

Wednesday 18th February at 2:00pm in the WWSR.
